Author's response to reviews

Title: The Skiers Knee without Swelling or Instability - A Difficult Diagnosis.

Authors:

Mark E O'Donnell (modonnell904@hotmail.com)
Stephen A Badger (stephen@badger.tc)
David Campbell (David.Campbell@bch.n-i.nhs.uk)
Willie Loan (wloan@doctors.org.uk)
Brendan Sinnott (brendan.sinnott@bch.n-i.nhs.uk)

Version: 3 Date: 10 January 2007

Author's response to reviews:

Dear Ms Claire Shaw,

Re: Submission of Revised Manuscript.

"The Skiers Knee without Swelling or Instability - A Difficult Diagnosis".

Please find enclosed the amendments as requested by email on the 8th January 2007.

All authors confirm that there are no competing interests with this manuscript.

The authors contributions are listed below;

Conception and design: MD, SB.
Analysis and interpretation: MD, SB, DC.
Data collection: MD, DC.
Writing the article: MD, SB, BS.
Critical revision of the article: MD, SB, DC, WL, BS.
Final approval of the article: MD, SB, WL, BS.
Statistical analysis: N/A.
Obtained funding: N/A.
Overall responsibility: MD.

Full informed consent was obtained from the patient.

I have also amended the title of the manuscript as requested.

If you require any further information please do not hesitate to contact me.

Yours Faithfully,
Mark O’Donnell DSEM MB(Dist) MFSEM(UK) MRCSEd.